PEER REVIEW HISTORY

BMJ Open publishes all reviews undertaken for accepted manuscripts. Reviewers are asked to complete a checklist review form (http://bmjopen.bmj.com/site/about/resources/checklist.pdf) and are provided with free text boxes to elaborate on their assessment. These free text comments are reproduced below.

ARTICLE DETAILS

TITLE (PROVISIONAL)	Can obesity and physical activity predict outcomes of elective knee or hip surgery due to osteoarthritis? – A meta-analysis of cohort studies.
AUTHORS	Pozzobon, Daniel; Ferreira, Paulo; Blyth, Fiona; Machado, Gustavo; Ferreira, Manuela

VERSION 1 – REVIEW

REVIEWER	Ethan Kuperman University of Iowa
	United States
REVIEW RETURNED	26-Jun-2017

INLAILM INLIGITION	20 0011 2017
GENERAL COMMENTS	This is a well-written and thorough examination of the impact of obesity on elective arthroplasty outcomes. It adds to the current literature and would be useful within the orthopedic community. Addressing the following, generally minor, concerns might somewhat improve the paper: 1. The abstract does not provide clarification on the magnitude of the effect of obesity on pain, disability, and postsurgical complications. 2. I do not see a figure 1 in the attached files. 3. The number of significant figures displayed is too high on the I2 values (we do not need 3 decimal points for this number in the figures) and P values < 0.001 are typically not displayed as 0.000 even if very small (as in the short- and long-term pain differences). 4. The values of the standardized mean differences for most of the outcomes reported, while statistically significant, were of a generally small magnitude (most < 0.5). The authors interpreted these as both clinically and statistically significant results. Would it be possible to demonstrate, in terms of an absolute percentage of patients (or some other absolute quantity) the clinical significance of these findings? This would enhance the strength of the current conclusions. 5. I don't see where the results from the first sentence, second paragraph of discussion are derived: (e.g. 70% symptom improvement at 6 months). If this comes from an earlier work by the same group, the citation is missing.

	6. The comparison of the # of studies with Liu et al in the discussion is somewhat misleadingthat team was only looking at hip arthroplasty, which may explain why this combined knee and hip study had a larger N. (in the "strengths and weaknesses in relation to other studies" paragraph of the discussion).
REVIEWER	Olalekan Uthman
	University of Warwick, UK
REVIEW RETURNED	19-Jul-2017

analysis was well performed and up to standard. However, my only concern was similar recent systematic reviews on the topic. This be for why this additional meta-analysis was conducted. Though, the authors tried to justify why their meta-analysis is more	GENERAL COMMENTS		9
--	------------------	--	---

REVIEWER	Nick Clement Freeman Arthroplasty Unit Newcastle Upon Tyne
	UK
REVIEW RETURNED	17-Aug-2017

GENERAL COMMENTS	The authors should congratulate on an excellent study. I have read this paper several times and have enjoyed it; learning something new each time. I am unable, despite my best efforts, to find any major flaws to prevent publication in the Journal.
	There is only two small points I would like to highlight for amendment:
	In the methods the authors state an exact methodology for identification of the literature through specific search engines. Then in the results the authors state they found 26 additional studies from elsewhere. This sounds like they have included them by choice. I feel either these 26 are not included or amend the methods to allow inclusion by whatever search criteria they were found. Second: tables 4 and 5 – Ollivier reference has 20012 should be
	2012

VERSION 1 – AUTHOR RESPONSE

Reviewer #1

1. The abstract does not provide clarification on the magnitude of the effect of obesity on pain, disability, and post-surgical complications.

Response: We have edited the abstract and added the requested information.

Results: 63 full papers were included in this systematic review. From these, 31 were included in the meta-analyses.

Our meta-analysis showed that non-obese participants tended to suffer less pain at both short (SMD - 0.43; 95%CI: -0.67 to -0.19; p<0.001) and long term (SMD -0.36; 95%CI: -0.47 to -0.24; p<0.001), less disability at long term (SMD -0.32; 95%CI: -0.36 to -0.28; p<0.001) and report fewer post-surgical complications at short (OR: 0.48; 95% CI: 0.25 to 0.91; p<0.001) and long term (OR: 0.55; 95% CI: 0.41 to 0.74; p<0.001) and less post-surgical infections after hip arthroplasty (OR: 0.33; 95% CI: 0.18 to 0.59; p<0.001), and particularly when compared to morbidly obese participants after knee arthroplasty (OR: 0.42; 95% CI: 0.23 to 0.78; p=0.006).

2. I do not see a figure 1 in the attached files.

Response: We apologise for the missing figure. We will make sure all the relevant appendixes files are correctly uploaded on the submission process of the reviewed version.

3. The number of significant figures displayed is too high on the I2 values (we do not need 3 decimal points for this number in the figures) and P values < 0.001 are typically not displayed as 0.000 even if very small (as in the short- and long-term pain differences).

Response: We have accepted the reviewer's suggestion and changed the decimals in all figures to just 2 decimal points.

4. The values of the standardized mean differences for most of the outcomes reported, while statistically significant, were of a generally small magnitude (most < 0.5). The authors interpreted these as both clinically and statistically significant results. Would it be possible to demonstrate, in terms of an absolute percentage of patients (or some other absolute quantity) the clinical significance of these findings? This would enhance the strength of the current conclusions.

Response: We accept the suggestion and added a sentence to the data analysis section to describe how we quantified the standardized mean differences on our results.

Page 8 (paragraph 1)

Between-study heterogeneity was calculated using I2 (I2 <25%: small heterogeneity; 25% 75%: large heterogeneity).(19) We have defined a standardized mean difference (SMD) smaller than 0.2 as a clinically small difference, between 0.3 and 0.7 as clinically moderate difference and greater than 0.8 as a clinically large difference.(20)

5. I don't see where the results from the first sentence, second paragraph of discussion are derived: (e.g. 70% symptom improvement at 6 months). If this comes from an earlier work by the same group, the citation is missing.

Response: We apologise for the lack of clarity. These results derive from our fractional polynomial analyses. The information has been added to the results section, as follows:

Page 21 (paragraph 2)

Our results from the fractional polynomial analysis have also shown that patients experience a favorable course of pain and disability post-surgery, with decreases in symptoms from baseline of approximately 70% at 6 months and 75% at 12 months for pain and 55% at 12 months and 67% at 120 months for disability.

6. The comparison of the # of studies with Liu et al in the discussion is somewhat misleading--that team was only looking at hip arthroplasty, which may explain why this combined knee and hip study had a larger N. (in the "strengths and weaknesses in relation to other studies" paragraph of the discussion).

Response: We have altered the discussion to better address the differences between previous studies and ours.

Page 21

Strengths and weaknesses in relation to other studies

Our meta-analysis results regarding the association between obesity and post-surgical complications found that obese patients present higher complication rates than non-obese patients. These results are consistent with the findings of previous systematic reviews of Hofstede,(21) Samson(22) and Liu.(23) Our meta-analysis results regarding the association between obesity and post-surgery disability also agreed with the findings of Buirs et al(24) and Samson et al(22) which found that obesity (defined as having BMI over 30 kg/m2), was associated with worst postsurgical functional score. The only previous review which has performed a meta-analysis on the association between obesity and post arthroplasty pain or disability limited its inclusion criteria to hip joint. (23) That review included a total of 15 studies in their meta-analysis and found that obesity increases the risk of postsurgical complications (RR: 1.68, 95% CI 1.23 to 2.30, P = 0.0004) and is associated with worse disability scores following surgery (MD: -2.75, 95% CI -4.77 to -0.6; P = 0.07). Our study has included 33 cohorts of hip arthroplasty participants in the qualitative analysis, 16 in the meta-analyses, and confirms past findings that obesity is associated with worse outcomes in terms of not only disability and complications, but also pain at both short and long term periods following surgery. Hofstede et al(21) have also conducted a systematic review of the literature on pre-operative predictors of surgical outcomes after hip replacement in patients with osteoarthritis. Although those authors included 35 studies, only 5 studies investigated the effect of obesity on post-surgical pain, disability and quality of life.(21) No meta-analysis was performed.

Reviewer #2

The manuscript reads well. The systematic review and meta-analysis was well performed and up to standard. However, my only concern was similar recent systematic reviews on the topic. This beg for why this additional meta-analysis was conducted. Though, the authors tried to justify why their meta-analysis is more comprehensive in the discussion section.

Reviewer #3

There is only two small points I would like to highlight for amendment:

In the methods the authors state an exact methodology for identification of the literature through specific search engines. Then in the results the authors state they found 26 additional studies from elsewhere. This sounds like they have included them by choice. I feel either these 26 are not included or amend the methods to allow inclusion by whatever search criteria they were found. Response: We have clarified the methods we used for identification of the literature (page 5, paragraph 2) and the result of these studies on the screening process (page 8, paragraph 3):

Page 5, paragraph 2

Data sources and searches

The first screening of potentially relevant records was conducted by one author (DP) based on titles and abstract, and two authors (DP and GM) independently performed the final selection of included trials based on full-text evaluation. A third reviewer arbitrated in case of disagreement (MF).

Response: Moreover, the reference lists of included studies were checked for potential studies. An additional 26 references were screened, none of which met our inclusion criteria. No restriction was applied on language.

Page 8, paragraph 4

Results

Our search strategy identified 11,990 studies. Twenty-six additional studies were identified through other sources and were included for screening. After removing 381 duplicates, 11,220 studies were screened and excluded based on keywords, titles, and abstracts. All the remaining 389 studies were written in English and were assessed by reading the full text, of which 327 were excluded, yielding 62 studies included in the systematic review. From these, 31 presented enough data to be included in at least one of the meta-analyses (Figure 1).

Second: tables 4 and 5 – Ollivier reference has 20012 should be 2012 Pages 18 and 19

Response: The typos on the tables 4 and 5 were corrected.

References

- 1. Pinheiro MB, Ferreira ML, Refshauge K, Ordonana JR, Machado GC, Prado LR, et al. Symptoms of Depression and Risk of New Episodes of Low Back Pain: A Systematic Review and Meta-Analysis. Arthritis Care Res (Hoboken). 2015;67(11):1591-603.
- 2. Beckenkamp PR, Lin C-WC, Chagpar S, Herbert RD, Ploeg HPvd, Moseley AM. Prognosis of Physical Function Following Ankle Fracture: A Systematic ReviewWith Meta-analysis. Journal of Orthopaedic & Sports Physical Therapy. 2014;44(11):841-51.
- 3. Wells G, Shea B, O'Connell D, Peterson J, Welch V, Losos M, et al. The Newcastle-Ottawa Scale (NOS) for assessing the quality of nonrandomised studies in meta-analyses. [cited 2017 29/08/17]. Available from: http://www.ohri.ca/programs/clinical_epidemiology/oxford.asp.
- 4. Higgins J, Green S, (editors). Cochrane Handbook for Systematic Reviews of Interventions Version 5.1.0 [updated March 2011]. The Cochrane Collaboration; 2011. Available from: http://handbook.cochrane.org.
- 5. Sedgwick P. Meta-analyses: how to read a funnel plot. BMJ: British Medical Journal. 2013;346.
- 6. Amin AK, Patton JT, Cook RE, Brenkel IJ. Does obesity influence the clinical outcome at five years following total knee replacement for osteoarthritis? Journal of Bone & Joint Surgery, British Volume. 2006 A;88B(3):335-40.
- 7. Collins RA, Walmsley PJ, Amin AK, Brenkel IJ, Clayton RAE. Does obesity influence clinical outcome at nine years following total knee replacement? Journal of Bone and Joint Surgery Series B. 2012;94 B(10):1351-5.
- 8. Font-Vizcarra L, Tornero E, Bori G, Bosch J, Mensa J, Soriano A. Relationship between intraoperative cultures during hip arthroplasty, obesity, and the risk of early prosthetic joint infection: A prospective study of 428 patients. International Journal of Artificial Organs. 2011;34(9):870-5.
- 9. Ibrahim T, Hobson S, Beiri A, Esler CN. No influence of body mass index on early outcome following total hip arthroplasty. International Orthopaedics. 2005;29(6):359-61.
- 10. Luebbeke A, Stern R, Garavaglia G, Zurcher L, Hoffmeyer P. Differences in outcomes of obese women and men undergoing primary total hip arthroplasty. Arthritis & Rheumatism-Arthritis Care & Research. 2007 A;57(2):327-34.
- 11. Maisongrosse P, Lepage B, Cavaignac E, Pailhe R, Reina N, Chiron P, et al. Obesity is no longer a risk factor for dislocation after total hip arthroplasty with a double-mobility cup. International Orthopaedics. 2014.

- 12. Amin AK, Clayton RA, Patton JT, Gaston M, Cook RE, Brenkel IJ. Total knee replacement in morbidly obese patients. Results of a prospective, matched study. Journal of Bone & Joint Surgery, British Volume. 2006 B;88B(10):1321-6.
- 13. Chee YH, Teoh KH, Sabnis BM, Ballantyne JA, Brenkel IJ. Total hip replacement in morbidly obese patients with osteoarthritis: RESULTS OF A PROSPECTIVELY MATCHED STUDY. Journal of Bone & Joint Surgery, British Volume. 2010;92(8):1066-71.
- 14. Dowsey MM, Liew D, Stoney JD, Choong PF. The impact of obesity on weight change and outcomes at 12 months in patients undergoing total hip arthroplasty. Medical Journal of Australia. 2010;193(1):17-21.
- 15. Kandil A, Werner BC, Gwathmey WF, Browne JA. Obesity, Morbid Obesity and their Related Medical Comorbidities are Associated with Increased Complications and Revision Rates after Unicompartmental Knee Arthroplasty. Journal of Arthroplasty. 2015;30(3):456-60.
- 16. Madsen AA, Taylor BC, Dimitris C, Hansen DC, Steensen RA, Gaines ST. Safety of bilateral total knee arthroplasty in morbidly obese patients. Orthopedics. 2014;37(3):e252-9.
- 17. Namba RS, Paxton L, Fithian DC, Stone ML. Obesity and perioperative morbidity in total hip and total knee arthroplasty patients. Journal of Arthroplasty. 2005;20(SUPPL. 3):46-50.
- 18. Napier RJ, O'Brien S, Bennett P, Doran E, Sykes A, Murray J, et al. Intra-operative and short term outcome of total knee arthroplasty in morbidly obese patients. Knee. 2014;21(3):784-8.
- 19. Higgins JPT, Thompson SG, Deeks JJ, Altman DG. Measuring inconsistency in meta-analyses. BMJ: British Medical Journal. 2003;327(7414):557-60.
- 20. Cohen J. Statistical power analysis of the behavioral sciences: New York: Academic Press; 1988.
- 21. Hofstede SN, Gademan MGJ, Vliet Vlieland TPM, Nelissen RGHH, Marang-van de Mheen PJ. Preoperative predictors for outcomes after total hip replacement in patients with osteoarthritis: a systematic review. BMC Musculoskeletal Disorders. 2016;17(1):212.
- 22. Samson AJ, Mercer GE, Campbell DG. Total knee replacement in the morbidly obese: a literature review. ANZ J Surg. 2010;80(9):595-9.
- 23. Liu W, Wahafu T, Cheng M, Cheng T, Zhang Y, Zhang X. The influence of obesity on primary total hip arthroplasty outcomes: A meta-analysis of prospective cohort studies. Orthopaedics & Traumatology: Surgery & Research. 2015;101(3):289-96.
- 24. Buirs LD, Van Beers LW, Scholtes VA, Pastoors T, Sprague S, Poolman RW. Predictors of physical functioning after total hip arthroplasty: a systematic review. BMJ Open. 2016;6(9):e010725.

VERSION 2 – REVIEW

REVIEWER	Ethan Kuperman
	University of Iowa Carver College of Medicine
	United States
REVIEW RETURNED	11-Oct-2017

GENERAL COMMENTS	My previous concerns have been addressed to my satisfaction.
	While there are some minor typos (e.g. First bullet under strengths
	and limitations: "themost" should be separated into two words) I
	believe this manuscript is nearly ready for publication.

REVIEWER	Nick Clement
	Orthopaedics and Trauma
	Freeman Hospital
	Newcastle
	UK
REVIEW RETURNED	19-Oct-2017

GENERAL COMMENTS	The authors should be congratulated on the quality of their work. This is a well written manuscript.
	The subject of paper is highly topical at the moment and would be of interest to the wider readership.
	My only suggestion or question is: would it be possible to look at change in score? It is well recognised that the post op score is
	worse compared to non-obese patients. However, the overall improvement is often equal to non-obese patients. Just though this
	might help give the reader insight, and not simply assume joint replacement should not be done in the obese. As far as am aware
	their is no literature showing a satisfaction difference according to weight. Hence, it is likely no the absolute post op score but the
	change relative to pre op function that is the most important factor.

VERSION 2 – AUTHOR RESPONSE

Reviewer: 1

My previous concerns have been addressed to my satisfaction. While there are some minor typos (e.g. First bullet under strengths and limitations: "themost" should be separated into two words) I believe this manuscript is nearly ready for publication.

Response: The manuscript was thoroughly read by the leading author and typos corrected.

Reviewer: 3

Reviewer Name: Nick Clement

The authors should be congratulated on the quality of their work. This is a well written manuscript.

Comment: The subject of paper is highly topical at the moment and would be of interest to the wider readership. My only suggestion or question is: would it be possible to look at change in score? It is well recognised that the post op score is worse compared to non-obese patients. However, the overall improvement is often equal to non-obese patients. Just though this might help give the reader insight, and not simply assume joint replacement should not be done in the obese.

As far as am aware there is no literature showing a satisfaction difference according to weight. Hence, it is likely no the absolute post op score but the change relative to pre op function that is the most important factor.

Response: We appreciate and thank the reviewer's suggestion, unfortunately we cannot present results based on change scores, as included studies did not provide measures of variability (e.g. SD) for change scores. These are necessary for the pooled analyses. We have added the following sentence to page 21, discussion section:

"We also highlight that although non-obese patients experience further improvements in pain and disability compared to obese participants, both groups improved significantly following surgery as depicted in figure 2."

VERSION 3 – REVIEW

REVIEWER	Nicholas Clement
	Department of Orthopaedics
	Royal Infirmary of Edinburgh
	UK
REVIEW RETURNED	22-Nov-2017

GENERAL COMMENTS	I congratulate the authors on their work in a very topical area at the
	moment.